Dear Chief Editor

We have made revisions and amended our manuscript accordingly based on the Reviewers opinions. Please see the revised document for more details. We appreciate your help in handling this paper and would be happy if this article can be published in the journal of BMC psychiatry in the near future. We look forward to hear from you soon.

Yours sincerely

Shen-Ing Liu
MD, PhD
Department of Psychiatry, Mackay Memorial Hospital
45 Min-sheng Road, Tam-Shui, New Taipei City, Taiwan
Telephone: 00-886-28094661 ext 3055
Fax: 00886-28094661 ext 3056

Email: maryliuyip@gmail.com; jennycyw@ntu.edu.tw